Author’s response to reviews

Title: MRI pattern suggests a new pathophysiological hypothesis for osteoarticular infection of the hip.

Authors:

Laura Merlini (laura.merlini@hcuge.ch)
Mehrak Anooshiravani (meharak.dumont@hcuge.ch)
Dimitri Ceroni (dimitri.ceroni@hcuge.ch)

Version: 2 Date: 16 June 2014

Author’s response to reviews: see over
Geneva 16/6/2014

Dear Editor,

we made the requested changes:

1) Title page is added with the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author
2) Competing Interests we included the statement “The authors declare that they have no competing interests.”
3.) Authors Contributions: we placed an Authors' Contributions section after the Competing Interests section.
4.) Consent Statement. This is a retrospective study, performed using MRI performed in a clinical every-day practice: parent informally approved the use (anonymized) for scientifical purposes.
5.) Acknowledgements We do not have acknowledgements.
6.) Line Numbering. We revised our manuscript including line numbers.

Thank you for your patience and Best Regards

Laura Merlini